Article details: 2022-0103

Title: Derivation and validation of pragmatic clinical models to predict hospital length of stay after cardiac surgery in Ontario, Canada

Authors: Alexandra Fottinger MBChB, Anan Bader Eddeen MSc, Douglas S. Lee MD PhD, Graham Woodward MSc, Louise Y. Sun MD SM

Reviewer 1

General comments (author response in bold)

This is manuscript that describes the development of two predictive models for LOS for CVSx patients.

1. The authors do not clearly describe how it can be applied clinically.

The reviewer raises an important point. We have added the following to describe the clinical applications of these models.

Page 13 line 6: "Operative decision-making, performance benchmarking and postoperative resource planning may be enhanced by objective tools... The ability to predict continuous hospital LOS is advantageous to hospital administrators, patients and caregivers. Patients often ask about their projected LOS as it sets realistic expectations of the costs and benefits of surgery as a part of the informed consent and therapeutic decision-making process. For caregivers, an estimated LOS will more effectively facilitate personal and professional arrangements in order to care for loved ones after surgery. At the hospital level, the widening use of electronic medical records may lend itself to institution-based self quality assessments, and to rationally allocate telemonitoring and other postdischarge care and follow-up resources to reduce morbidity and readmissions. At the system level, these models could be used to generate risk-adjusted observed vs. expected LOS to systematically evaluate performance."

Page 14 line 6: "The ability to identify those at risk for extremely prolonged LOS allows for better decision-making from the perspectives of the healthcare system as well as the individual patient. As the system level, this ability, coupled with continuous LOS prediction, could facilitate data-driven clinical scheduling to increase throughput, facilitate targeted interventions such as prehabilitation, Enhanced Recovery After Surgery, and early referral to continuing care facilities. As prolonged LOS has also been implicated with increased healthcare cost(1) and disability after discharge,(2–6) our predictive models will inform effective provider-patient discussions and encourage patient-centered operative decision-making."

Page 15 line 4: "Importantly, our ability to predict continuous LOS enables precision-based hospital capacity planning, as well as ... incentivized allocation of healthcare funding. Incorporation of the model into tools such as the provincewide CorHealth information system could also help individual providers to understand bed requirements at the time of intervention, allowing for more accurate resource planning."

2. The authors fail to adequately contextualize how their model compares to existing models (they only compare their model to another one...)

The majority of cardiac surgery literature on post-operative LOS centers on LOS in the CVICU and not of the entire hospital stay. There is in fact very sparse literature regarding total duration of hospital LOS in cardiac surgical patients, the

most robust and commonly used being the repurposing of STS and EuroScore for predicting hospital LOS as a binary variable. Accordingly, we had described these single center validation studies of the EuroScore to predict prolonged hospital LOS >12 days (7) and of the STS score to predict the probability of "short" and "long" LOS, with 14 days as the cutoff (8,9). We had described the potential uses (please refer to our response to your Comment #1) and advantages of our models over the STS and EuroScore, including their ability to identify top-tier resource users while also projecting continuous LOS with excellent accuracy. We have made revisions in the manuscript to highlight this fact. Page 6, line 11: "Nonetheless, while risk factors have been identified for prolonged postoperative hospital LOS, very few models are available to predict this important metric, and none are able to estimate actual post-operative LOS with accuracy. Further, though existing models include those from the Society of Thoracic Surgeons (STS) and the EuroSCORE datasets (1,7,9), they were developed to predict perioperative mortality and end organ morbidity and were only later validated in single center datasets for the purpose of predicting prolonged LOS in a binary fashion and are not used to estimate actual LOS." Page 14, line 16: "Comparatively, the EuroScore had a c-statistic of 0.71 (0.69-0.72) for predicting prolonged hospital LOS (> 12 days) when validated in a monocentric setting(7), and the STS model had a c-statistic of 0.716-0.732 for predicting short hospital LOS of \leq 5 days and 0.739-0.796 for predicting prolonged stay of > 5 days, depending on the type of surgery performed.(10)"

3. Not certain whether the readership of CMAJ would benefit from this study. Perhaps a more surgical-specific journal should be targeted.

The CMAJ showcases a variety of medical and surgical research, including many recent studies that specifically pertain to cardiac surgery (5,11–15). Given the potential impact of the present study at the level of patients, caregivers, hospitals and the healthcare system, we believe our findings will be of interest to readers of CMAJ.

4. The models also appear to add precious little in terms of novelty.

There are very few models that predict hospital LOS after cardiac surgery, and none that predict continuous LOS with accuracy. Therefore, the novelty of our study lies in its ability to address the need to accurately plan hospital and postdischarge resources via an efficient and data-driven approach to managing the mounting surgical backlog.

5. On a separate note, it would be easier to follow the manuscript if it was formatted properly and consistently. There are paragraphs where the between-line spacing is double (?) whereas the rest are not.

The manuscript has been reformatted.

Reviewer 2: Susan Brogly PhD MSc

Affiliation: Queen's Department of Surgery; Kingston Health Science Centre, Surgery General comments (author response in bold)

The authors present findings on a model to predict hospital length of stay (LOS) after cardiac surgery in patients 18 and older in Ontario, Canada. It is stated that the models improve upon those previously published due to the population-based cohort in which

the model was developed. It is unclear who would use the models and for what purpose. These concerns and some additional suggestions are further described below.

Abstract

1. Perhaps type of resource intensity should be qualified; non-surgical personnel? surgeon? cost to health care provider? OR time?

We appreciate the comment and have amended the abstract to note the multidisciplinary care required to facilitate discharge. Certainly while the intraoperative course, operative time, and surgical personnel are an important component of cardiac surgical care, this manuscript aims to describe the postoperative length of stay and as such references the resources required for appropriate post-operative care).

Abstract, Page 4, line 2: "Cardiac surgery is resource intensive and often requires considerable length of stay (LOS) and multidisciplinary involvement to facilitate discharge."

The model fit predictors are not the key findings of interest here, rather it is the characteristics that predict longer length of stay. These should be added.
We thank the reviewer for this suggestion and have incorporated key characteristics predicting longer LOS in the abstract as suggested.
Abstract, Page 4, line 15: "Factors predicting prolonged LOS included age, female sex, procedure type and urgency, comorbidities including frailty, high-risk acute coronary syndrome, heart failure, reduced left ventricular ejection fraction, psychiatric and pulmonary circulatory disease."

3. How will the models benchmark quality, and quality of what? In addition, it is unclear how the model will inform the patient-centred decisions. Most patients will want cardiac surgery to prevent mortality regardless of LOS. If surgery allotments, OR time, cardiac surgeon availability, costs etc. are of concern to the hospital administration and Ministry of Health, then the models may be of use to them to determine which patients will undergo surgery. This should be clear in the objective and implications. We appreciate this suggestion and have revised the abstract to reflect the idea of benchmarking clinical performance based on expected length of stay. Abstract, Page 4, line 21: "Our models could be used to benchmark quality based on expected length of stay"

Methods

4. It would be clearer to start out that the cohort was identified using administrative health data held at ICES in the single-payer healthcare system in Ontario, Canada. And then go on to describe the privacy of ICES.

We have revised as suggested, by adding on page 7, line 7: "Ontario operates a publicly funded, universal health care system that reimbursed all covered services and clinicians.", as well as moving the description of privacy at ICES after this statement.

5. Need to specify that index procedure means "first cardiac surgery"; could just use the latter and avoid term index altogether.

We have revised as the reviewer suggested.

6. Was there some reason that surgeon characteristics: age, sex, years of practice, number of cardiac surgeries (and/or specific types per year) were not included? It seems these may be useful predictors.

The models were developed to predict hospital LOS at the time of surgical triaging, often before surgeons are assigned to the case. Thus, provider information is not always available at the time the models are applied.

7. Need to define how LOS was calculated: from date of cardiac surgery to discharge or from date of admission? The former would be more accurate given that there may be inter-hospital transfers etc.

Thank you for the opportunity to clarify our methodology. LOS is calculated from the date of surgery until the date of discharge from index hospital.

Page 9 line 7: "The primary outcome was postoperative hospital LOS, from the date of surgery until date of index hospital discharge."

8. Unclear why 98th percentile (35 days) was selected for the outliers. Is there a clinical reason? A statistical one? Why not the 90th?

Thank you for the opportunity to clarify our methodology. We have added the following to explain the rationale for choosing the 98th percentile as cut-off for the binary prediction model.

Abstract: "To address the rightward skew in LOS data and to identify top-tier resource users, we used logistic regression to derive a model to predict the likelihood of LOS being >98th percentile (≥35 days)."

Methods, page 9 line 8: "We derived two separate models: the first (binary outcome model) to identify the top-tier resource users (i.e., LOS exceeding the 98th percentile value of \geq 35 days), and the other (continuous outcome model) to predict the actual LOS in days in the remainder of the cohort. The rationale for our modeling approach was two-fold. First, as LOS data is invariably right-skewed with extreme values in those with prolonged stay, (16,17) the choice of 98th percentile (LOS \geq 35 days) as cut-off eliminates extreme values and increases the precision of continuous LOS prediction. Second, those with LOS \geq 35 days are amongst the highest resource users and are most likely to acquire a new disability that would impair their quality of life after eventual discharge. (2–6) The ability to predict extremely prolonged LOS will allow patients to be informed partners in the therapeutic decision-making process"

Interpretation, Page 14 line 6: "The ability to identify those at risk for extremely prolonged LOS allows for better decision-making from the perspectives of the healthcare system as well as the individual patient. As the system level, this ability, coupled with continuous LOS prediction, could facilitate data-driven clinical scheduling to increase throughput, facilitate targeted interventions such as prehabilitation, Enhanced Recovery After Surgery, and early referral to continuing care facilities. As prolonged LOS has also been implicated with increased healthcare cost(1) and disability after discharge,(2–6) our predictive models will inform effective provider-patient discussions and encourage patient-centered operative decision-making."

9. Terminology should be consistent throughout the manuscript: sometimes actual is used for the model of all patients and sometimes continuous. It would be easier to follow with consistent terminology.

Thank you for pointing this out. We have replaced "actual" with "continuous" throughout, for consistency and clarity.

10. Why were variables modeled as continuous? Is this clinically relevant? **We have added the following to clarify.**

Page 13 line 6: "Operative decision-making, performance benchmarking and postoperative resource planning may be enhanced by objective tools... The ability to predict continuous hospital LOS is advantageous to hospital administrators, patients and caregivers. Patients often ask about their projected LOS as it sets realistic expectations of the costs and benefits of surgery as a part of the informed consent and therapeutic decision-making process. For caregivers, an estimated LOS will more effectively facilitate personal and professional arrangements in order to care for loved ones after surgery. At the hospital level, the widening use of electronic medical records may lend itself to institution-based self quality assessments, and to rationally allocate telemonitoring and other postdischarge care and follow-up resources to reduce morbidity and readmissions. At the system level, these models could be used to generate risk-adjusted observed vs. expected LOS to systematically evaluate performance." Page 15 line 4: "Importantly, our ability to predict continuous LOS enables precision-based hospital capacity planning, as well as ... incentivized allocation of

healthcare funding. Incorporation of the model into tools such as the provincewide CorHealth information system could also help individual providers to understand bed requirements at the time of intervention, allowing for more accurate resource planning."

Results

11. While it is appreciated that the authors highlight differences in patients in Table 1, some key features are missed. For instance, a minority of the population is female, and the age difference in groups is 6 years etc.

Thank you for the comment. Amendments to the manuscript have been made to better reflect the demographic differences between the two populations described in Table 1.

Page 11, line 13: "A minority of the patients in either group were female, though females more likely than males to have prolonged hospital LOS. Those with prolonged hospital LOS were more likely to be older, with a 6 year median age difference noted, of lower income levels, and were more likely to present urgently and emergently for complex procedures (redo sternotomy, CABG + valve(s), multiple valves and thoracic aorta surgery) of longer surgical duration and at teaching hospitals. They are also more likely to have a higher multimorbidity burden including frailty, in addition to cardiovascular comorbidities such as recent myocardial infarction, reduced LVEF, and higher NYHA classification."

12. Since study size is large—and the p-value is a reflection of the difference and the study size—suggest replacing the p-values in Table 1 with standardized differences. Thank you for this suggestion. We have updated Table 1 accordingly.

Discussion

13. Unclear to this reader how this model would be used. There did not appear to be an equation in the manuscript—nor a link to an on-line calculator—to allow the cardiac surgeon (or hospital administration) to calculate predicted LOS. Who is the model for? Thank you for this suggestion. We are currently working on digital risk calculators and will have them ready prior to publication.

14. Likely irrelevant to the cardiac surgeon (though he/she will monitor patient while in recovery) and more relevant to hospital administration who will manage beds, costs, and support staff. Related to this is "The ability to identify those at risk for prolonged LOS allows for better decision-making from the healthcare system as well as the individual patient" is not transparent. What aspect of the healthcare system?

Thank you for the opportunity to expand on our discussion. We have provided clarification below:

Page 13, line 19: "At the system level, these models could be used to generate risk-adjusted observed vs. expected LOS to systematically evaluate performance."

Page 14, line 10: "As prolonged LOS has also been implicated with increased healthcare cost(1) and disability after discharge,(2–6) our predictive models will inform effective provider-patient discussions and encourage patient-centered operative decision-making.

Page 16, line 10: "Care, outcomes, and patient satisfaction may be substantially improved if clinical judgment is supported by objective quantification in the planning of care. Shared decision-making through a thorough discussion of evidence-based estimates of risk and benefits of treatment options with patients and caregivers, is vital to patient-centered care. Our research fills an important knowledge gap by providing scientific evidence to inform shared clinical decisionmaking."

15. Also, as mentioned, LOS is less important to the patient than undergoing needed cardiac surgery.

We have added the following to clarify our point about the importance of LOS prediction for patients, caregivers and administrators:

Page 13 line 11: "The ability to predict continuous hospital LOS is advantageous to hospital administrators, patients and caregivers. Patients often ask about their projected LOS as it sets realistic expectations of the costs and benefits of surgery as a part of the informed consent and therapeutic decision-making process. For caregivers, an estimated LOS will more effectively facilitate personal and professional arrangements in order to care for loved ones after surgery. At the hospital level, the widening use of electronic medical records may lend itself to institution-based self quality assessments, and to rationally allocate telemonitoring and other post-discharge care and follow-up resources to reduce morbidity and readmissions. At the system level, these models could be used to generate risk-adjusted observed vs. expected LOS to systematically evaluate performance."

16. Unclear which models relied on designated staff for the variables needed; prior papers? present manuscript? It seems that the current model also would require staff to identify the variables. Later it is stated that the variables included in the model are routinely collected; perhaps in ICES but the ICES data are unavailable to hospital

administrators and certainly the surgeons do not have time to collate these data from their patient charts.

The models reported in the present study are based on routinely collected, data from a standardized, province-wide surgery referral form that is completed by the referring physician. Information from this referral form is directly available to the triaging team at each Ontario cardiac centre. In comparison, the STS models make predictions with 50 covariates. The collection of STS data requires additional infrastructure to be in place, which is not practical for the Canadian healthcare system.(13)

17. Unclear that a model for all cardiac surgeries is an advantage over models developed in homogenous groups of cardiac surgery patients.

We agree that modeling could be performed either in an omnibus fashion across all procedure types, or within a subgroup of patients who underwent a particular type of surgical procedure. The advantage of omnibus models is that they are very practical and are widely applicable across all types of cardiac surgeries, and that predictions could still be tailored to the complexity of the procedure. On the other hand, as there are many potential combinations of complex cardiac procedures, models that are specific to a single procedure type are not readily applicable to patients who undergo a variant of that procedure (i.e., an AVR-only or AVR+CABG model is not applicable to patients who undergo AVR + MAZE). It is likely for these reasons that the widely used models in cardiac surgery are omnibus.

REFERENCES

1. Osnabrugge RL, Speir AM, Head SJ, Jones PG, Ailawadi G, Fonner CE, et al. Prediction of costs and length of stay in coronary artery bypass grafting. Ann Thorac Surg. 2014 Oct;98(4):1286–93.

2. Diab MS, Bilkhu R, Soppa G, Edsell M, Fletcher N, Heiberg J, et al. The influence of prolonged intensive care stay on quality of life, recovery, and clinical outcomes following cardiac surgery: A prospective cohort study. J Thorac Cardiovasc Surg. 2018 Nov;156(5):1906-1915.e3.

3. Sun LY, Spence SD, Benton S, Beanlands RS, Austin PC, Bader Eddeen A, et al. Age, Not Sex, Modifies the Effect of Frailty on Long-term Outcomes After Cardiac Surgery. Ann Surg. 2020 Jun;

4. Tran DTT, Tu J V., Dupuis JY, Eddeen AB, Sun LY. Association of frailty and long-term survival in patients undergoing coronary artery bypass grafting. J Am Heart Assoc. 2018;

5. Sun LY, Eddeen AB, Mesana TG. Disability-free survival after major cardiac surgery: a population-based retrospective cohort study. C open. 2021;9(2):E384–93.

6. Sun LY, Rodger J, Duffett L, Tulloch H, Crean AM, Chong A-Y, et al. Derivation of Patient-Defined Adverse Cardiovascular and Noncardiovascular Events Through a Modified Delphi Process. JAMA Netw open. 2021 Jan;4(1):e2032095.

7. Toumpoulis IK, Anagnostopoulos CE, Swistel DG, DeRose Jr JJ. Does EuroSCORE predict length of stay and specific postoperative complications after cardiac surgery? Eur J Cardio-Thoracic Surg [Internet]. 2005 Jan 1;27(1):128–33. Available from: https://doi.org/10.1016/j.ejcts.2004.09.020 8. Triana AJ, Vyas R, Shah AS, Tiwari V. Predicting Length of Stay of Coronary Artery Bypass Grafting Patients Using Machine Learning. J Surg Res. 2021 Mar;264:68–75.

9. Peterson ED, Coombs LP, Ferguson TB, Shroyer AL, DeLong ER, Grover FL, et al. Hospital variability in length of stay after coronary artery bypass surgery: results from the Society of Thoracic Surgeon's National Cardiac Database. Ann Thorac Surg. 2002 Aug;74(2):464–73.

10. O'Brien SM, Feng L, He X, Xian Y, Jacobs JP, Badhwar V, et al. The Society of Thoracic Surgeons 2018 Adult Cardiac Surgery Risk Models: Part 2-Statistical Methods and Results. Ann Thorac Surg. 2018 May;105(5):1419–28.

11. de Waard D, Fagan A, Minnaar Ć, Horne D. Management of patients after coronary artery bypass grafting surgery: a guide for primary care practitioners. C Can Med Assoc J = J l'Association medicale Can. 2021 May;193(19):E689–94.

12. Sun LY, Wijeysundera HC, Lee DS, van Diepen S, Ruel M, Eddeen AB, et al. Derivation and validation of a clinical risk score to predict death among patients awaiting cardiac surgery in Ontario, Canada: a population-based study. C open. 2022;10(1):E173–82.

13. Sun LY, Chu A, Tam DY, Wang X, Fang J, Austin PC, et al. Derivation and validation of predictive indices for 30-day mortality after coronary and valvular surgery in Ontario, Canada. C Can Med Assoc J = J I'Association medicale Can. 2021 Nov;193(46):E1757–65.

14. Sun LY, Eddeen AB, Wijeysundera HC, Mamas MA, Tam DY, Mesana TG. Derivation and validation of a clinical model to predict death or cardiac hospitalizations while on the cardiac surgery waitlist. C Can Med Assoc J = J l'Association medicale Can. 2021 Aug;193(34):E1333–40.

15. Wang J, Vahid S, Eberg M, Milroy S, Milkovich J, Wright FC, et al. Clearing the surgical backlog caused by COVID-19 in Ontario: a time series modelling study. C Can Med Assoc J = J l'Association medicale Can. 2020 Nov;192(44):E1347–56.

16. Austin PC, Rothwell DM, Tu J V. A Comparison of Statistical Modeling Strategies for Analyzing Length of Stay after CABG Surgery. Heal Serv Outcomes Res Methodol [Internet]. 2002;3(2):107–33. Available from: https://doi.org/10.1023/A:1024260023851

17. Almashrafi A, Elmontsri M, Aylin P. Systematic review of factors influencing length of stay in ICU after adult cardiac surgery. BMC Health Serv Res [Internet]. 2016;16(1):318. Available from: https://doi.org/10.1186/s12913-016-1591-3

18. T. TDT, V. TJ, Jean-Yves D, Anan BE, Y. SL. Association of Frailty and Long-Term Survival in Patients Undergoing Coronary Artery Bypass Grafting. J Am Heart Assoc [Internet]. 2018 Aug 7;7(15):e009882. Available from: https://doi.org/10.1161/JAHA.118.009882

19. Sun LY, Gaudino M, Chen RJ, Bader Eddeen A, Ruel M. Long-term Outcomes in Patients With Severely Reduced Left Ventricular Ejection Fraction Undergoing Percutaneous Coronary Intervention vs Coronary Artery Bypass Grafting. JAMA Cardiol. 2020 Jun;5(6):631–41.

20. Public Hospitals Act. Regulation 964 - Amended to O. Reg. 321/01.

21. Gilbert T, Neuburger J, Kraindler J, Keeble E, Smith P, Ariti C, et al. Development and validation of a Hospital Frailty Risk Score focusing on older people in acute care settings using electronic hospital records: an observational study. Lancet (London, England). 2018 May;391(10132):1775–82. 22. Meyer M, Schwarz T, Renkawitz T, Maderbacher G, Grifka J, Weber M. Hospital Frailty Risk Score predicts adverse events in revision total hip and knee arthroplasty. Int Orthop. 2021 Nov;45(11):2765–72.

23. Ontario Ministry of Health and Long Term Care. Health Analyst's Toolkit. [Internet]. [cited 2015 Nov 14]. Available from:

http://www.health.gov.on.ca/english/providers/pub/healthanalytics/health_toolkit/h ealth_toolkit.pdf.

24. du Plessis V, Beshiri R, Bollman RD CH. Definitions of "rural". Agricultural and Rural Working Paper Series, No. 61. 2002.

25. Sun LY, Tu J V, Sherrard H, Rodger N, Coutinho T, Turek M, et al. Sex-Specific Trends in Incidence and Mortality for Urban and Rural Ambulatory Patients with Heart Failure in Eastern Ontario from 1994 to 2013. J Card Fail. 2018 Sep;24(9):568–74.

26. Prina AM, Cosco TD, Dening T, Beekman A, Brayne C, Huisman M. The association between depressive symptoms in the community, non-psychiatric hospital admission and hospital outcomes: a systematic review. J Psychosom Res. 2015 Jan;78(1):25–33.

27. Fisher K, Griffith LE, Gruneir A, Kanters D, Markle-Reid M, Ploeg J. Functional limitations in people with multimorbidity and the association with mental health conditions: Baseline data from the Canadian Longitudinal Study on Aging (CLSA). PLoS One. 2021;16(8):e0255907.

28. Oakes G. Report on Adult Cardiac Surgery: Isolated Coronary Artery Bypass Graft (CABG) Surgery, Isolated Aortic Valve Replacement (AVR) Surgery and Combined CABG and AVR Surgery October 2011 - March 2016. Toronto; 2018.

29. Panah M, Andres LA, Strope SA, Vela-Cantos F, Bennett-Guerrero E. Morbidity associated with prolonged hospital lenght of stay following cardiac surgery. Anesth Analg [Internet]. 1998;86(4S). Available from: https://journals.lww.com/anesthesia-

analgesia/Fulltext/1998/04001/MORBIDITY_ASSOCIATED_WITH_PROLONGED_HO SPITAL.50.aspx

30. Nakano M, Nomura Y, Suffredini G, Bush B, Tian J, Yamaguchi A, et al. Functional Outcomes of Frail Patients After Cardiac Surgery: An Observational Study. Anesth Analg [Internet]. 2020;130(6). Available from: https://journals.lww.com/anesthesia-

analgesia/Fulltext/2020/06000/Functional_Outcomes_of_Frail_Patients_After.12.as px

31. Sun LY, Tu J V, Lee DS, Beanlands RS, Ruel M, Austin PC, et al. Disabilityfree survival after coronary artery bypass grafting in women and men with heart failure. Open Hear. 2018;5(2):e000911.